Physics 115A Spring 2001 Statistical Physics due 12:30 pm Thursday, June 7

PROBLEM SET 9 Final Exam: The final exam will take place on Tuesday June 12 from 10:30 am to 12:30 pm in 108 Rowland Hall. This is the officially scheduled time and place. The final will cover chapters 1 to 7, chapter 9, and section 8.5. You are also responsible for all the material covered in lecture (see lecture notes on web). You may bring one 8.5” x 11” sheet of paper to the exam. Write whatever you think may help you on this piece of paper. You may write on both sides. (Ian, Mitchell and Matt will take the final on Friday June 15 in PSCB 210.) Review Session: A review session will be held on Monday June 11 in 108 Rowland Hall from 11:00 – 11:50 am.

Reading: Chapter 9 in Reif. You may skip section 9.12

1. Reif 9.1

2. Reif 9.2

3. Reif 9.3

4. Reif 9.4

5. Reif 9.9

6. Reif 9.13